Check for updates

OPEN ACCESS

EDITED BY Morteza Arab-Zozani, Birjand University of Medical Sciences, Iran

REVIEWED BY Alvin Ang, Ateneo de Manila University, Philippines Xufeng Cui, Zhongnan University of Economics and Law, China Chao Zheng, Shandong University, China

*CORRESPONDENCE Zhengxiu Sun ⊠ sunzhengxiu1130@126.com

RECEIVED 17 October 2023 ACCEPTED 09 January 2024 PUBLISHED 02 February 2024

CITATION

Huang P, Sun Z, Li L and Li J (2024) Has the Integrated Medical Insurance System promoted return-to-hometown entrepreneurship among migrant workers? Evidence from China. *Front. Public Health* 12:1323359. doi: 10.3389/fpubh.2024.1323359

COPYRIGHT

© 2024 Huang, Sun, Li and Li. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms.

Has the Integrated Medical Insurance System promoted return-to-hometown entrepreneurship among migrant workers? Evidence from China

Pengcheng Huang¹, Zhengxiu Sun^{2*}, Linfang Li³ and Jia Li⁴

¹School of Finance, Anhui University of Finance and Economics, Bengbu, China, ²School of Economics and Management, Southeast University, Nanjing, China, ³Jinan No.1 High School in Shandong Province, Jinan, China, ⁴School of Clinical Medicine, Chengdu University of Traditional Chinese Medicine, Chengdu, China

An important way to reduce urban-rural disparity lies in encouraging migrant workers to return to their hometowns for entrepreneurship. This paper examines the effect of the Integrated Medical Insurance System on the return-to-hometown entrepreneurship among migrant workers. Using microdata from the China Household Finance Survey (CHFS) spanning from 2013 to 2019, we find that the Integrated Medical Insurance System (IMIS) significantly increases the likelihood of migrant workers returning to their hometowns for entrepreneurship by 0.44%. This result remains stable after a series of robustness checks. Heterogeneity results indicate that this "pullback effect" is more pronounced for those who are male and with lower educational levels, higher income, larger social networks, and lower risk preferences. Finally, the interaction between the Mass Entrepreneurship and Innovation policy (MEI) and IMIS can create a more significant combined effect in promoting the return of migrant workers to their hometowns for entrepreneurial activities.

KEYWORDS

Integrated Medical Insurance System, return-to-hometown entrepreneurship, migrant workers, pullback effects, urban-rural mobility

1 Introduction

In recent decades, developing countries, including China, have witnessed a pronounced trend of rural-to-urban migration. Migrant workers hold a crucial position in advancing economic development and constitute a central segment of the mobile population (1). The movement of these labor forces from rural areas has given rise to the phenomenon commonly referred to as "rural hollowing," which has led to a dearth of rural labor, an aging demographic, and an increase in poverty levels (2). This phenomenon has contributed to the deterioration of rural regions and the further exacerbation of the urban–rural divide. Therefore, facilitating the return of migrant workers from their jobs abroad to their hometowns for work and life seems to be a potential solution.

The impact of welfare programs on population mobility and employment has garnered substantial attention from economists and policymakers (3). In general, individuals are more inclined to migrate to regions offering generous welfare benefits, a phenomenon commonly

referred to as the "welfare magnet effect" (4-7). Medical insurance, as a crucial facet of social welfare programs, plays a pivotal role in alleviating population mobility stemming from employment-related factors, often referred to as "job lock" (8-12). It is noteworthy that these findings primarily originate from empirical studies conducted in developed countries, with limited evidence available for developing countries. Among the scarce studies based on Chinese cases, Shi (13) discovered that a new rural health insurance program initiated by the Chinese government in 2003, known as the New Cooperative Rural Medical Scheme (NCRMS), had negative effects on rural-to-urban migration mobility in China. Similarly, Wang et al. (3) revealed that the NCRMS reduced the likelihood of rural residents working in other regions but did not observe a significant "pullback effect" for migrant workers returning to their hometowns or nearby regions from elsewhere. However, it is crucial to recognize that these outcomes are indicative of the previous NCRMS, which, while effectively curbing the outflow of rural migrant workers through its residency-based healthcare insurance system, also impedes the return of laborers engaged in external employment. This situation is possibly detrimental to the goal of narrowing urban-rural disparities.

In 2016, the State Council of China issued the "Opinions on Integrating the Basic Medical Insurance Systems for Urban and Rural Residents," with the overarching aim of reducing healthcare coverage disparities between urban and rural areas. This was achieved by merging the NCRMS and the Urban Resident Basic Medical Insurance (URBMI) into the Urban and Rural Resident Basic Medical Insurance (URBMI). Compared to the previous NCRMS, the URRBMI offers higher reimbursement rates, expanded reimbursement ceilings, and a more streamlined reimbursement procedure, thereby significantly narrowing the disparity in healthcare benefits between urban and rural areas (14). This development raises an intriguing scientific inquiry: Can this plausibly exogenous policy alteration stimulate the return of rural laborers who work outside their hometowns? To the best of our knowledge, there is currently no existing literature that offers a more rigorous causal inference on this particular issue.

Several local governments across China have progressively undertaken the integration of URBMI and NCMS, culminating in the creation of the Integrated Medical Insurance System (IMIS). This presents a unique opportunity to investigate this matter within a Difference-in-Differences (DID) framework. Thus, we leverage data from the China Household Finance Survey (CHFS) spanning from the years 2013 to 2019, employing a DID approach to assess the impact of IMIS on migrant workers' return to their hometowns for entrepreneurial pursuits. Our DID results reveal that IMIS significantly increases the likelihood of migrant workers returning to their hometowns for entrepreneurship by 0.44%. This finding remains robust following a series of rigorous tests, including DIDM, endogeneity, alternative models, and placebo tests. Heterogeneity results suggest that the pullback effect after the implementation of IMIS is more pronounced for those who are male and with lower educational levels, higher income, larger social networks, and lower risk preferences. Considering that the entrepreneurship policy, Mass Entrepreneurship and Innovation (MEI), coexists within the timeframe of our study, and after excluding its potential contamination on the outcomes, our moderating effect model shows that the interaction between the MEI and IMIS yields a more substantial combined impact in facilitating the return of migrant workers to their hometowns for entrepreneurial ventures.

We make the following marginal contributions: Firstly, our findings enrich the existing body of research on the policy impacts of the IMIS. Prior studies have provided valuable insights into the impact of IMIS, including its role in reducing overwork among migrant workers (15), enhancing the well-being of rural older adults (16), improving healthcare access and quality while lowering medical expenses (17), alleviating medical impoverishment (18), promoting healthcare equity (19-21), enhancing health status (17, 22), fostering social integration (23), and boosting employment (24, 25), as well as increasing the utilization rate of healthcare services among the rural migrant population (19, 26-28). However, in contrast to prior research focused on the impact of healthcare insurance on employment and health welfare, our study extends its scope to the entrepreneurial decisions of migrant workers. This expansion provides a novel theoretical framework for comprehending the dynamics of rural labor force. Our findings complement this existing body of work by highlighting its role in facilitating the return of migrant workers to their hometowns to initiate entrepreneurial ventures and observing heterogeneous responses within different subgroups. Furthermore, it is worth noting that the integration of URBMI and NCRMS into IMIS represents an enhancement of the social security system, effectively addressing several shortcomings of the previous NCRMS. Our findings align with the broader literature on the "welfare magnet effect" and "pullback effect," where regions offering enhanced welfare benefits tend to attract population inflows (29, 30).

Secondly, our study makes a significant contribution to the field of entrepreneurship determinants. Prior studies have extensively explored the multitude of factors that influence individuals' decisions to embark on entrepreneurial ventures, including individual characteristics (31, 32), household attributes (33, 34), the prevailing economic landscape (35, 36), and the cultural milieu (37). However, it is worth highlighting that, to the best of our knowledge, there is a relative scarcity of studies that offer more rigorous causal inferences from the perspective of the health insurance system. Our study is an important supplement to this field.

Thirdly, our findings hold significant implications for policymaking and practice. They underscore the potential of improving healthcare insurance system in promoting the return of migrant workers to their hometowns for entrepreneurial endeavors. Additionally, our study reveals a valuable policy synergy between healthcare coverage and policies aimed at fostering rural entrepreneurship. To some extent, this could reflect the great potential of optimizing healthcare insurance in addressing pressing issues such as rural depopulation, the urban–rural divide, the enhancement of rural healthcare welfare, and the promotion of balanced economic development. Consequently, for countries grappling with similar challenges involving disparities in urban–rural healthcare welfare and the outflow of rural populations, China's experience may offer a feasible solution.

The remainder of the paper is organized as follows: Section 2 introduces the IMIS and migrant workers' return-to-hometown entrepreneurship. Section 3 presents the empirical design. Section 4 describes the data sources, variable definitions, and summary statistics. Section 5 reports the benchmark result, a series of robustness tests, heterogeneous results, and moderating effects. Section 6 offers a discussion of our findings. Section 7 are conclusion and policy implications.

TABLE 1 The IMIS implementation in provinces (or municipalities).

Provinces (or municipalities)	Year
Tianjin, Chongqing	2009
Ningxia	2010
Guangdong	2012
Shandong	2014
Shanghai	2015
Fujian, Zhejiang, Hainan	2016
Yunnan, Inner Mongolia, Jilin, Sichuan, Anhui, Shanxi, Guangxi, Jiangxi, Hebei, Henan, Hubei, Hunan, Qinghai, Gansu, Xinjiang, Shaanxi	2017
Heilongjiang, Beijing, Jiangsu, Guizhou	2018
Liaoning	2019
Tibet	2020

The data of IMIS implementation year, collected by authors, is derived from the official website of each provincial or municipal government.

2 Institutional background

2.1 IMIS

The IMIS in China aims to rectify disparities in healthcare coverage between urban and rural areas, with the objective of enhancing the equity and sustainability of universal healthcare coverage. In the past, this system encountered several significant challenges. First, urban residents enjoyed more comprehensive medical insurance coverage and higher reimbursement rates compared to their rural counterparts (38, 39). Second, there was a lack of uniformity between urban and rural medical insurance systems, resulting in substantial disparities in healthcare benefits (40). Third, the funding sources for rural residents' medical insurance were unstable and often faced financial shortfalls (41, 42). To address these issues, the Chinese government initiated the process of implementing the IMIS. In 2003, the National Health and Family Planning Commission introduced the policy of establishing a multi-level medical security mechanism that covered both urban and rural areas (43). On January 12, 2016, the State Council issued the "Opinions on Integrating the Basic Medical Insurance System for Urban and Rural Residents," announcing large-scale efforts to standardize urban and rural medical insurance systems nationwide.1 Subsequently, various provinces introduced corresponding implementation rules and policy measures to facilitate the rollout of the IMIS. Specific implementation timelines for each province (or municipality) are outlined in Table 1.

After the implementation of the IMIS, the situation of residents in both rural areas and urban areas improved significantly, and healthcare coverage was significantly enhanced in different dimensions (44). Firstly, the reimbursement rate was raised. For instance, in Shanghai, the hospitalization reimbursement rate was increased to 70% after the implementation of IMIS in 2017, compared to the previous rates of 60 and 50% for URBMI and NRCMS, respectively. In Beijing, both outpatient and inpatient reimbursement rates increased by 5% in 2018.² In Shanxi Province, after the 2017 IMIS, the average reimbursement rate for total hospitalization expenses within the medical insurance catalog saw a 15% increase.3 Secondly, the list of reimbursed drugs was expanded. For instance, in Hebei Province, the original NRCMS covered approximately 1,000 types of drugs, while URBMI covered around 2,400 types. After the 2017 integration, the URRBMI expanded to cover approximately 2,900 types of drugs.⁴ In Yunnan Province, following the 2017 IMIS, the number of reimbursed drugs for rural residents increased by 1,476 types, and urban residents witnessed an increase of 442 types. This expansion of the drug catalog was more substantial for rural residents.⁵ Thirdly, a greater number of designated medical institutions were included. For instance, in Beijing, the number of designated medical institutions expanded from over 800 under NRCMS and more than 2,000 under URBMI to over 3,000 in total. In Shanxi Province, the number of designated medical and pharmaceutical institutions grew from over 2,000 to over 7,000.

However, there are still issues of fairness in the integration of this system. On one hand, the tiered payment and benefits do not truly reflect fairness. For instance, in Zhejiang Province, the urban-rural medical insurance is designed with multiple tiers, associating low contributions with low benefits and high contributions with high benefits (45). On the other hand, there is a lack of fairness in the reimbursement for offsite medical treatments. Due to the inconsistency in the scope of medical institutions providing diagnosis, treatment, and reimbursement services, patients seeking treatment in different areas may face difficulties in reimbursement based on the variations in qualifications among medical institutions. This diminishes the efficiency of healthcare insurance services and exacerbates regional disparities. Furthermore, disparities between local healthcare insurance policies and reimbursement systems hinder the settlement of healthcare insurance for offsite treatments, leading to regional disparities that affect the convenience of medical treatment for the public and increase the complexity of overall management (46).

2.2 Migrant workers' return-to-hometown entrepreneurship

The trend of return-to-hometown entrepreneurship among Chinese migrant workers is becoming increasingly evident. According to statistics, the number of people who return to rural hometown for entrepreneurship exceeded 8.5 million in 2019,⁶ and it is projected to surpass 15 million nationwide by 2025.⁷ This phenomenon can be attributed to two main factors. On the one hand, there is the "push" factor from urban areas. This is evident in three ways. Firstly, China's unique household registration (*hukou*) policy has exacerbated social exclusion of migrant workers (47). Secondly, the escalating cost of living results in heightened socioeconomic pressure on them. Thirdly, the number of jobs suitable for migrant workers is limited (48, 49). On

¹ https://www.gov.cn/gongbao/content/2016/content_5036268.htm

² http://bj.bendibao.com/zffw/20171214/246632.shtm

³ http://www.gov.cn/xinwen/2017-11/12/content_5239129.htm

⁴ http://www.chinanews.com/gn/2016/10-10/8025768.shtml

⁵ http://www.gov.cn/xinwen/2017-12/19/content_5248363.htm

⁶ https://www.gov.cn/xinwen/2019-12/16/content_5461646.htm

⁷ https://www.gov.cn/xinwen/2020-07/17/content_5527709.htm

the other hand, there is the "pull" factor in rural areas. This is evident in two ways. Firstly, improvements in rural infrastructure, including living conditions, transportation networks, and connectivity, may attract rural residents to return to their hometowns for entrepreneurship (50). Secondly, the Chinese Government has consistently emphasized the promotion of entrepreneurship among migrant workers who return to their hometowns to encourage the return of labor forces. For example, in 2014, the government recommended facilitating the transfer of rural migrant workers' employment to nearby areas and providing support for them to return to their hometowns to start their own businesses.⁸

The success of migrant workers' return-to-hometown entrepreneurship is of significant importance for the government to formulate sound entrepreneurship and rural development policies. Previous studies have provided evidence that the entrepreneurship of migrant workers returning to their hometowns, as a manifestation of the return of rural labor, complements rural human capital and contributes to the advancement of rural economic development (51). Return-to-hometown entrepreneurs possess advantages in terms of economic, human, and social capital, enabling them to swiftly seize market opportunities and leverage the diverse resource endowments in their local areas. This helps effectively rectify China's longstanding pattern of one-way flow of high-quality capital from rural to urban areas (52). Moreover, it plays a crucial role in driving rural industrial transformation, upgrading, reducing the urban-rural development gap, contributing to the integrated development of urban and rural areas, and employment promotion (53, 54).

In summary, there may be an important connection between IMIS and the phenomenon of migrant workers' return-to-hometown entrepreneurship. With the implementation of IMIS, not only has it provided greater security and stability in terms of healthcare for migrant workers, but it has also promoted urban-rural integration on a broader scale, creating more favorable conditions for their returnto-hometown entrepreneurship. This integration represents not only a significant advancement in healthcare but also a key driver for advancing socioeconomic development in rural areas. By improving health insurance coverage, IMIS boosts the confidence of migrant workers in returning to their hometowns for entrepreneurship. Furthermore, the experiences and skills they have accumulated in urban areas can be applied in rural regions, promoting diversification and innovation in local economies. Therefore, in-depth research on the impact of IMIS on the entrepreneurship of migrant workers returning to their hometowns not only holds theoretical significance but also has important practical value for guiding policy practices and promoting balanced urban-rural development.

3 Empirical strategies

3.1 Benchmark model

The traditional difference-in-differences (DID) model is applicable only when the timing of policy implementation is consistent. Given the uneven implementation timing of IMIS across various regions, this study follows the approach outlined by Beck et al. (55) and establishes a time-varying DID model to evaluate the impact of IMIS on migrant workers' return-to-hometown entrepreneurship. The specification of the time-varying DID model is as follows:

$$RE_{ipt} = \beta_0 + \beta_1 IMIS_{ipt} + \gamma X_{ipt} + \tau_p + \omega_t + \varepsilon_{ipt}$$
(1)

where RE_{ipt} is the outcome variable, indicating whether the rural migrant worker *i* in the province *p* return to their hometowns to start an entrepreneurship in year *t*. *IMIS*_{*ipt*} is the URRBMI integration conditions, which indicates whether IMIS was implemented in year *t* in the province *p* where individual *i* was located. β_1 is the coefficient of interest, which measures the causal impact of IMIS on return-to-hometown entrepreneurship among migrant workers. X_{ipt} is a list of individual-level and household-level controls. τ_p denotes province fixed effects, which absorbs the impact of time-invariant province characteristics. ω_t denotes year fixed effects, which captures changes over time that affect respondents in different provinces similarly. ε_{ipt} is the error term.

4 Data, variables and summary statistic

4.1 Data

The micro-level data for this study are sourced from the China Household Finance Survey (CHFS) conducted in four waves in the years 2013, 2015, 2017, and 2019. Initiated by the China Household Finance Survey and Research Center at Southwestern University of Finance and Economics in 2011, the CHFS is conducted biennially and has been updated to include data up to 2019. The most recent survey covers a sample distribution across 29 provinces (autonomous regions or municipalities directly under the central government), 367 counties (districts or county-level cities), and 1,481 communities, encompassing 40,011 households and 127,012 individuals. This survey collects data at three hierarchical levels: individual, household, and regional. It includes comprehensive information on population characteristics, employment status, mobility, housing assets, financial wealth, income, consumption, social security, insurance, and other relevant factors, rendering it well-suited for the purposes of our study. This wealth of information in the questionnaire enables us to select a multitude of control variables, thereby minimizing the potential for endogeneity bias due to omitted variables. Additionally, the four waves of longitudinal survey data align precisely with the primary implementation period of the IMIS.

We perform the following data preprocessing steps: Firstly, as the implementation of IMIS primarily affected individuals with agricultural *hukou*, and return-to-hometown entrepreneurship was most prevalent among the working-age population, we filter the CHFS database to include only individuals with agricultural *hukou* aged

⁸ For instance, "Opinions on Promoting High-Quality Development of Returning to Hometown and Entrepreneurship," jointly issued by the National Development and Reform Commission and 19 other departments in 2020, and "Opinions on Ensuring the Current Employment and Entrepreneurship of Migrant Workers," jointly issued by the Ministry of Human Resources and Social Security and the National Development and Reform Commission, along with 15 other departments.

between 18 and 65. Secondly, in the 2017 survey wave of the CHFS database, the question used to assess residents' "risk preferences" was posed exclusively to new households, leading to missing values for this variable in the tracking sample. To address this, we impute missing values using responses from the original households in 2015.⁹ Thirdly, the timing of IMIS implementation varies among different prefecture-level cities within individual provinces. Since the micro-level database only allows for province-level identification, we exclude provinces where there is inconsistency in policy implementation timing among prefecture-level cities within the same province. In the end, we have a total of 50,559 valid samples.

4.2 Definitions of variables

Outcome variable: the outcome variable is *Return-to-hometown entrepreneurship* (RE), which is a dummy variable that equals 1 if an individual has returned to their hometown and is involved in work categorized as "self-employed, operating an individual or private business, or running an online store"¹⁰ during the survey year, and 0 if otherwise.

Independent variables: the independent variable is *IMIS*. We determine the implementation time of IMIS in various prefecture-level cities based on detailed implementation documents. It is a dummy variable which equals 1 if the individual's respondence takes place before the implementation of IMIS in his/her province, and 0 if otherwise.

Control variables: given the high relevance of individual and household-level factors to migrant workers' decisions to engage in entrepreneurship upon returning to their hometowns, following Shi et al. (56), Guo and Wang (57), and Chen and Zhang (58), we incorporate control variables at individual and household levels. Specifically, individual-level variables include *Age*, *Age*², *Gender*, *Education level, Marital status, Annual income, Self-reported health, Pension insurance, Medical insurance*, and *Risk preference. Age* is defined as the difference between the individual's survey wave and

year of birth. Considering the nonlinear effect of age on migrant workers' return-to-hometown entrepreneurship, we include a squared term for age, i.e., Age.² Gender is a dummy variable which equals 1 if the individual is male, and 0 if otherwise. Educational level is defined on a corresponding scale from 0 to 8, with values assigned based on the individual's highest reported educational level, ranging from illiteracy (0) to primary school (1), junior high school (2), senior high school (3), vocational school (4), associate's degrees (5), bachelor's degrees (6), master's degrees (7), and doctoral degrees (8).¹¹ Marital status is a dummy variable which equals 1 if the individual is married, cohabiting, or remarried, and 0 if otherwise. Annual income is the logarithm of all the income an individual earns in a year, including salary, bonuses, money management, etc. Based on an individual's response to the question, "Compared to people of your age, how would you rate your current physical health?," Self-reported health is a dummy variable which equals 0 if the response is poor or very poor, and 1 if otherwise. Social pension insurance is a dummy variable which equals 1 if the individual has any form of pension insurance,¹² and 0 if otherwise. Social medical insurance is a dummy variable which equals 1 if the individual has any form of medical insurance,¹³ and 0 if otherwise. Based on the response to the question, "If you have a sum of money for investment, which type of investment project would you prefer?," Risk preference takes a value of 0 for not willing to take any risks, 1 for slightly low risk, 2 for moderate risk, 3 for slightly high risk, and 4 for high risk. Household-level variables include Family size, Number of members under 18, and Household renging¹⁴ expenditures. Family size refers to the number of family members economically connected to the household. Number of members under 18 is the number of individuals in the household who are younger than 18 years old. Household renging expenditures represent the logarithm of cash or non-cash expenditures made by the household to individuals other than the parents of both spouses.

⁹ The rationale behind this approach is grounded in the general observation that an individual's risk preference reflects their personality traits, which typically do not undergo drastic changes. Hence, our hypothesis is predicated on the assumption that an individual's risk preference remains stable, barring any significant shocks. To validate the consistency of risk preference, we also compared the same individual's risk preferences in 2015 and 2019. The results indicate that only 824 samples exhibited changes in risk preference between 2015 and 2019, constituting approximately 3.8% of the 2015 sample size. Upon examination, it was found that excluding these samples does not materially affect the principal findings of this study.

¹⁰ The choice of "self-employed, operating an individual or private business, or running an online store" as proxies for entrepreneurship is justified in the survey by these options embodying the essential characteristics of entrepreneurship, including autonomy, innovation, and risk-taking. These roles typically involve full personal control and management of a business, entail certain operational risks, and are often pursued to achieve personal career goals and financial independence. Additionally, these activities reflect the entrepreneurial spirit, such as independent thinking, innovative capability, and self-motivation.

¹¹ The rationale for categorizing education level as an ordinal variable stems from two main considerations. Firstly, the dataset encompasses a spectrum of nine distinct options, spanning from illiteracy to doctoral degrees. Secondly, this classification into nine categories allows for a more nuanced reflection of individual differences in educational levels, thereby enabling a more effective control for potential impacts arising from diverse stages of educational experiences on the outcomes examined in this study.

¹² These pensions come from government or public institution retirement pensions.

¹³ These social insurance programs include Urban Employee Basic Medical Insurance (UEBMI), URBMI, NRCMS, URRBMI, and Free Medical Care (FMC).

¹⁴ *Renqing* is a concept of significant importance in Chinese culture regarding social interactions and interpersonal relationships. It typically refers to the acts of mutual assistance, care, respect, and politeness displayed by individuals in the context of social exchanges. These behaviors play a crucial role in establishing and maintaining positive interpersonal relationships. *Renqing* encompasses activities such as the exchange of gifts, hosting meals, the giving of symbolic presents, and offering assistance, all of which serve as expressions of gratitude, respect, and friendship.

4.3 Summary statistics

Table 2 presents sample characteristics based on the implementation of IMIS in the regions where individuals reside. It includes mean and standard deviation values for the full sample, the IMIS subsample, and the non-IMIS subsample, along with significance tests for differences in means between these two subsamples. The results in Table 2 reveal that from 2013 to 2019, the proportion of individuals engaged in return-to-hometown entrepreneurship among the working-age population remained relatively low, accounting for only 1.18%. Moreover, the IMIS subsample shows significantly higher levels of individual annual income and subjective well-being compared to the non-IMIS subsample. Additionally, the statistical results for various control variables suggest that individuals in the IMIS subsample, when compared to those in the non-IMIS subsample, exhibit characteristics such as older ages, higher educational attainment, lower marriage rates, higher income levels, better health status, a higher participation rate in social pension insurance, a lower participation rate in social medical insurance, lower risk preferences, fewer family members, and lower family renging expenditures.

5 Results

5.1 Common trends test

The main advantage of the time-varying DID identification strategy is its ability to control for unobserved confounders that change over time, which could otherwise bias our estimates. However, its validity relies on the parallel trends assumption, which posits that in the presence of time-fixed effects, province-fixed effects, and individual and household-level control factors, there should be no significant differences in the changes of the outcome variable across provinces prior to the policy shock. In other words, any potential confounders that are not directly observed in the data would follow a parallel path in both the treatment and control groups in the absence of the policy intervention. By comparing the differences in outcomes before and after the policy implementation across these groups, we can isolate the impact of the policy from these unobserved factors. Furthermore, this method enhances the credibility of our findings by allowing for a more nuanced analysis of the policy's impact. It enables us to observe how the effect of the policy unfolds over time, which is crucial in understanding the dynamics of policy implementation and its consequences.

The event study approach is employed in various ways to assess the impact of policies on outcome variables. First, it enables an examination of dynamic changes in outcome variables before and after policy implementation by defining relative time windows. Analyzing the pre-policy change trend helps ascertain whether the parallel trends assumption holds, while observing the post-policy dynamic trend provides insights into the policy's long-term effects. Second, the event study method contributes to more robust causal identification by accounting for a range of observable and unobservable confounding factors, thus enhancing the effectiveness of causal inference. Consequently, we estimate the following event study specification:

$$RE_{ipj} = \beta_0 + \sum_{j=-5}^{3} \beta_1 IMIS_{pj} + \gamma X_{ipj} + \tau_p + \omega_j + \varepsilon_{ipj}$$
(2)

where the subscript j is the relative period between the time when individual *i* receives the interview and the implementation of IMIS in

TABLE	2	Summary	statistics.
-------	---	---------	-------------

Variables	Total sample	IMIS subsample	Non-IMIS subsample	Difference		
Panel A: Outcome variables	Panel A: Outcome variables					
Return-to-hometown entrepreneurship	0.0118 (0.0004)	0.0168 (0.0007)	0.0048 (0.0004)	0.0120*** (0.0009)		
Panel B: Individual variable	S					
Age	39.9053 (0.0448)	40.2928 (0.0594)	39.3594 (0.0678)	0.9334*** (0.0908)		
Gender	0.6291 (0.0019)	0.6269 (0.0025)	0.6322 (0.0029)	-0.0053 (0.0038)		
Educational level	2.3440 (0.0054)	2.3847 (0.0072)	2.2868 (0.0083)	0.0978*** (0.0110)		
Marital status	0.8247 (0.0015)	0.8138 (0.0020)	0.8401 (0.0022)	-0.0263*** (0.0030)		
Annual income	10.8326 (0.0054)	10.9543 (0.0064)	10.6613 (0.0093)	0.2929*** (0.0109)		
Self-reported health	0.0830 (0.0011)	0.0865 (0.0014)	0.0780 (0.0016)	0.0085*** (0.0022)		
Social pension insurance	0.6406 (0.0019)	0.6529 (0.0024)	0.6232 (0.0029)	0.0297*** (0.0038)		
Social medical insurance	0.9299 (0.0010)	0.9265 (0.0013)	0.9346 (0.0015)	-0.0080*** (0.0020)		
Risk preference	0.8202 (0.0044)	0.7742 (0.0056)	0.8850 (0.0071)	-0.1107*** (0.0090)		
Panel C: Household-level variables						
Family size	4.2116 (0.1327)	4.1565 (0.0085)	4.2892 (0.0103)	-0.1327*** (0.0133)		
Number of members under 18	0.8297 (0.0035)	0.8281 (0.0047)	0.8319 (0.0053)	-0.0038 (0.0072)		
Household renqing expenditures	5.8288 (0.0139)	5.4597 (0.0189)	6.3487 (0.0200)	-0.8890*** (0.0280)		

Standard errors are shown in parentheses.

***, ** and * indicate the t-test significance levels of the difference in mean values between IMIS and non-IMIS subsamples.

The significance levels of 1, 5, and 10% are denoted by ***, **, and *, respectively.



his/her province. For instance, j=0 means that IMIS has been implemented in the current year in the province where individual i is located. The settings for other variables are the same as those in Equation (1).

Figure 1, based on Equation (2), presents the annual trend for RE. We utilize the year preceding IMIS implementation in each province as the reference point. Prior to IMIS, the coefficients of the interaction terms exhibit small absolute magnitudes and statistical insignificance. This suggests that there is no significant difference in the change of the outcome variable between provinces before the implementation of IMIS. After IMIS, the 95% confidence intervals (CIs) of the interaction term coefficients overlap, indicating that the common trends assumption holds for the outcome variable, namely, RE.

5.2 Benchmark results

Table 3 presents the benchmark results based on Equation (1). Specifically, in column (1), no control variables are included. Column (2) introduces individual-level control variables, while column (3) adds household-level control variables to those in column (2). Column (4) further includes time fixed effects and province fixed effects. The results demonstrate that the implementation of IMIS significantly enhances the likelihood of migrant workers returning to their hometowns for entrepreneurial endeavors. The magnitude of this effect ranges from 0.44 to 1.34%, and all estimated coefficients are statistically significant at the 1% level. We consider the results from column (4) as the benchmark, indicating that IMIS significantly increases the probability of migrant workers returning to their hometowns to start a business by 0.44%. This unintended finding holds significant practical implications for China, especially in the context of the ongoing implementation of the Rural Revitalization Strategy,15 as China has long grappled with issues related to rural labor outflow and shortages (59). The reflow of labor is expected to play a pivotal role in improving the distribution of labor between urban and rural areas, boosting rural economic vitality, narrowing the urban-rural gap, and establishing a unified national labor market (60).

5.3 Robustness checks

5.3.1 DIDM

Following the principles of staggered DID, and considering the varying timing of IMIS implementation across different regions, we are concerned that some groups that have already experienced the treatment may inadvertently be used as control groups in the traditional DID approach. This can result in the problem of negative weights, potentially biasing the policy effect (61). To investigate this hypothesis, we adopt the approach suggested by de Chaisemartin and D'Haultfoeuille (62) to address the negative weights issue, particularly when the Average Treatment Effects (ATEs) exhibit heterogeneity across groups or time periods. The method focuses on correctly identifying and employing control and treatment groups in the context of staggered policy implementation. It accounts for the heterogeneity in ATEs across different groups and time periods, which is a common occurrence in staggered policy settings. Specifically, we employ a DID_{g,t} estimator to assess the impact of IMIS on migrant workers' return-to-hometown entrepreneurship. By using a DID_{et} estimator, this approach allows for a more accurate assessment of the policy's impact, factoring in the varying timings of its implementation and the diverse effects it may have across different regions or groups. The $DID_{g,t}$ estimation is specified as Equation (3):

$$DID_{g,l} = E\left(Y_{g,F_g+l}\left(0_{F_g-1},1_{l+1}\right) - Y_{g,F_g+l}\left(0_{F_g+l}\right)\right)$$
(3)

where DID_{g,l} is the expected difference between the treated individual g's status about whether they return to their hometowns to start an entrepreneurship at $F_g + l$ (denoted by $Y_{g,F_g+l}(0_{F_g-1},1_{l+1})$) and the counterfactual "status quo" status for an individual would have obtained if an individual had remained untreated from period one to $F_g + l$ (denoted by $Y_{g,F_s+l}(0_{F_s+l})$). Y_{g,F_s+l} denotes an individual g's statues about whether they return to their hometowns to start an entrepreneurship at period $F_g + l$, and the treatment condition from period one to $F_{g} + l$ are defined in the brackets. F_{g} denotes the first period at which individual g's IMIS was implemented. l is the length of exposure to the treatment. 0_{F_g-1} and 0_{F_g+l} denote $1 \times (F_g-1)$ and $1 \times (F_g + l)$ vectors with coordinates equal to 0. Similarly, l_{l+1} denote a $1 \times (l+1)$ vector with coordinates equal to 1. $DID_{g,l}$ estimators are aggregated into a DID₁ estimator, to estimate the effect of IMIS on individuals' status about whether they return to their hometowns to start an entrepreneurship for l periods. As recommended by de Chaisemartin and D'Haultfœuille (62), this study clusters the standard errors to the individual level.

de Chaisemartin and D'Haultfœuille (62) shows DID_l estimator can be extended to conditional DID designs and is unbiased under conditional parallel trends assumption which requires request the migrant workers' return-to-hometown entrepreneurship in the region before the occurrence of the IMIS has the same trend. To test the parallel trends assumption, the placebo estimators are used corresponding to the ATEs estimated before the implementation of

¹⁵ The Rural Revitalization Strategy is a comprehensive initiative launched by the Chinese government in 2018, aimed at fostering economic development in rural areas, improving the living standards of rural residents, and achieving balanced development between urban and rural regions.

TABLE 3 Benchmark results.

Variable	(1)	(2)	(3)	(4)	(5)
IMIS	0.0120***	0.0128***	0.0134***	0.0044***	0.0295***
	(0.001)	(0.001)	(0.001)	(0.002)	(0.002)
Age		0.0024***	0.0022***	0.0021***	0.0024***
		(0.000)	(0.000)	(0.000)	(0.000)
Age2		-0.0000***	-0.0000***	-0.0000***	-0.0000***
		(0.000)	(0.000)	(0.000)	(0.000)
Gender		0.0031***	0.0031***	0.0031***	0.0041***
		(0.001)	(0.001)	(0.001)	(0.001)
Education level		0.0011***	0.0010***	0.0009***	0.0011**
		(0.000)	(0.000)	(0.000)	(0.001)
Marital status		0.0058***	0.0031***	0.0050***	0.0051***
		(0.001)	(0.001)	(0.001)	(0.002)
Self-reported health		-0.0031**	-0.0026*	-0.0033**	-0.0031
		(0.001)	(0.001)	(0.001)	(0.002)
Social pension insurance		-0.0024***	-0.0027***	-0.0022**	-0.0022
		(0.001)	(0.001)	(0.001)	(0.001)
Social medical insurance		0.0050***	0.0047***	0.0042***	0.0034
		(0.001)	(0.001)	(0.001)	(0.003)
Annual income		-0.0015***	-0.0016***	-0.0021***	-0.0028***
		(0.000)	(0.000)	(0.000)	(0.001)
Family size			-0.0008***	-0.0009***	-0.0016***
			(0.000)	(0.000)	(0.001)
Number of members under			0.0037***	0.0031***	0.0045***
18			(0.001)	(0.001)	(0.001)
Household <i>renqing</i>			0.0008***	0.0010***	0.0011***
expenditures			(0.000)	(0.000)	(0.000)
Risk preference			0.0013***	0.0020***	0.0027***
			(0.000)	(0.002)	(0.001)
Constant	0.0048***	-0.0399***	-0.0390***	-0.0436***	-0.0302***
	(0.000)	(0.007)	(0.007)	(0.007)	(0.011)
Time fixed effects	(0.000)	(0.007)	(0.007)		
Time fixed effects Province fixed effects				YES YES	YES YES
Observations	65,770	65,770	65,770	65,770	32,769
<i>R</i> -squared	0.003	0.006	0.007	0.015	0.017

Standard errors are clustered at the individual level.

***, **, and * denote significance levels of 1, 5, and 10%, respectively.

the IMIS. The DID_l and placebo estimators are computed by the did_multiplegt package for Stata (63).

Columns (1)–(3) of Table 4 present the results of the impact of IMIS on RE among migrant workers using the de Chaisemartin and D'Haultfœuille (62) approach. In column (1), without the inclusion of control variables, the estimated coefficient for IMIS stands at 0.0126, and it is statistically significant (p < 0.01). When we control for individual and household characteristics in column (2), the coefficient decreases to 0.0093 but remains statistically significant (p < 0.01). Notably, the results in columns (1) and (2) capture the transient effect

of IMIS, representing the immediate impact of the policy. To provide a more precise estimate of the policy effect, accounting for the postpolicy impact, we estimate the dynamic effect in column (3) and calculate the average policy effect of IMIS within our study waves. In column (3), the first and second rows of results suggest that the effect of IMIS in promoting migrant workers' return-to-hometown entrepreneurship gradually increases over time, consistent with the post-policy trend in Figure 1. The third row in column (3) represents the average effect of IMIS, while the fourth row demonstrates that all results of the placebo test in the pre-policy period, as seen in columns (1)–(3), are not statistically significant, aligning with the pre-policy

TABLE 4 The impact of IMIS on RE among migrant workers using DIDM.

Variable	(1) DIDM	(2) DIDM	(3) DIDM
IMISt	0.0126***	0.0093***	0.0079***
	(0.0036)	(0.0032)	(0.0028)
IMIS _{t+1}			0.0422***
			(0.0068)
Average			0.0110***
			(0.0027)
Placebo_1	-0.0139	-0.0156	0.0127
	(0.0162)	(0.0106)	(0.0095)
Control variables	No	Yes	Yes
Observations	65,770	65,770	65,770

Control variables are included at individual and household levels as shown in Table 3. Standard errors are clustered at the individual level. ***, **, and * denote significance levels of 1, 5, and 10%, respectively.



trend in Figure 1. Figure 2 visually illustrates the dynamic trends observed in column (3) of Table 4. This outcome implies that the policy effects resulting from IMIS implementation are not incidental, substantiating their significance.

5.3.2 Endogeneity

As presented in Table 1, the implementation of the IMIS across provinces exhibits a staggered pattern. It is essential to recognize that the timing of IMIS implementation in specific provinces may not have been a random occurrence but rather a result of policy considerations stemming from endogenous factors, including distinct local economic, demographic, and environmental conditions.

This raises doubts about the credibility of the baseline regression results in our study. China's work plan for IMIS mandates that each region should formulate specific implementation plans by the end of December 2016 and put them into action promptly. Consequently, when comparing cities that implemented IMIS in the vicinity of 2017 with those that did so in the year 2017, it becomes evident that the latter group was more likely to be influenced by the exogenous deadline stipulated by national policy. In other words, the provinces that implemented IMIS in 2017 are more conducive to aiding us in mitigating endogeneity concerns arising from autonomous decisions. In this section, we confine our analysis to provinces that adopted IMIS in the year 2017 and proceed to re-evaluate the policy impacts of IMIS in accordance with Equation (1). The result in column (5) of Table 3 indicates that the coefficient for IMIS remains significantly positive at the 1% level of significance, which substantially alleviates concerns about endogeneity issues in our findings.

5.3.3 Alternative model

In the benchmark model, we utilize OLS regression. However, since the outcome variable in this study, i.e., RE, is binary, and recognizing that different model specifications can potentially impact the results, in this section, we conduct Probit and Logit regressions on Equation (1). Probit¹⁶ and Logit¹⁷ models are both used in statistics for binary outcome modeling, where the response variable can take one of two possible outcomes. These models are particularly useful in situations where the outcome is a probability of an event occurring. The results of marginal effects presented in Table 5 demonstrate that the outcomes, following the model replacement, do not significantly deviate from the benchmark results. This indicates that our core findings remain robust across various model specifications.

5.3.4 Placebo test

Given the possibility that changes in trends for the treatment and control groups after IMIS implementation may be influenced by other policies or random factors, it becomes imperative to conduct a placebo test to ascertain whether IMIS was indeed the driving factor, thereby eliminating the potential contamination of results by other unobservable factors. In this section, we undertake a placebo test by randomizing the treatment and control groups. The specific procedure is as follows: Firstly, we maintain the original policy implementation timing unchanged. If n regions initiate IMIS implementation in year t, we randomly select n regions from all regions as those where IMIS commences in that year. At this stage, these n regions form the newly generated random treatment groups, while the remaining regions naturally become the newly generated random control groups. Subsequently, based on this assignment, we re-estimate Equation (1) using the newly generated random treatment and control groups and repeat this process 500 times. This results in 500 estimated coefficients for IMIS implementation.

The probability density distribution of the estimated coefficients from randomization is depicted in Figure 3. The results reveal that the distribution of estimated coefficients for IMIS closely approximates a normal distribution with a mean near 0. When we compare this distribution with the coefficient of 0.0044 for IMIS implementation in column (4) of Table 3, it becomes evident that the coefficient in the

¹⁶ The Probit model is based on the cumulative normal distribution function. It assumes that the underlying process generating the binary outcome follows a normal distribution. In the Probit model, the probability of the outcome variable equaling a particular value (usually 1) is modeled as a function of a normal cumulative distribution of the independent variables.

¹⁷ The Logit model, on the other hand, is based on the logistic function. It uses the logistic distribution to model the probability of the binary outcome. The logistic function in the Logit model transforms the linear combination of the independent variables, providing a probability that ranges between 0 and 1.

TABLE 5 Alternative model.

Variable	(1)	(2)	
	Probit	Logit	
IMIS	0.0048**	0.0050**	
	(0.0020)	(0.0022)	
Control variables	YES	YES	
Time fixed effects	YES	YES	
Province fixed effects	YES	YES	
Observations	65,770	65,770	
R-squared	0.1147	0.1162	

Control variables are included at individual and household levels as shown in Table 3. Standard errors are clustered at the individual level.

***, **, and * denote significance levels of 1, 5, and 10%, respectively.



baseline regression significantly differs from the coefficient in the placebo test. As a result, we can conclude that our baseline finding, indicating that IMIS promotes return-to-hometown entrepreneurship among rural migrant workers, is less likely to be influenced by other policies or unobservable random factors.

5.4 Heterogeneity analysis

Gender: within Panel A of Table 6, we furnish estimates stratified by gender. Findings suggest a positive yet statistically insignificant impact of IMIS on female migrant workers. In contrast, male migrant workers exhibit a more pronounced inclination to return to their hometowns for entrepreneurial endeavors, evidenced by a statistically significant effect of 0.0051 (p < 0.05). This disparity implies the presence of gender-based variations in the effectiveness of IMIS in fostering entrepreneurship that draws individuals back to their hometowns among migrant laborers. migrant workers over the age of 45, IMIS does not have a significant effect on their return to business. This indicates that IMIS has a relatively stronger influence on encouraging younger individuals to return to their hometowns for entrepreneurial activities.¹⁸

Education Level: we divide the total sample into two sub-samples based on educational levels, distinguishing between low and high education. Specifically, rural migrant workers with education levels of 'illiteracy,' 'primary school,' and 'junior high school' were classified as having low education, while others were categorized as having high education.¹⁹ Findings in Panel C of Table 6 reveal that, for migrant workers with lower education, IMIS significantly increases their probability of returning for entrepreneurial activities by 0.50% (p < 0.05). Conversely, for those with higher education, the effect of IMIS is positive but not statistically significant. This suggests that migrant workers with comparatively lower educational attainment are more inclined to return to their hometowns for entrepreneurship following the implementation of IMIS, consistent with the findings of Tamvada et al. (67).

Annual income: the total sample was divided based on median income, distinguishing between low-income (<median) and highincome (\geq median) migrant workers. The findings in Panel D of Table 6 reveal that IMIS significantly enhances the likelihood of return-to-hometown entrepreneurship for high-income migrant workers by 0.88% (p < 0.01). However, this effect is not statistically significant for low-income migrant workers. This outcome implies that IMIS exerts a more substantial incentive impact on migrant workers with relatively higher annual incomes compared to those with lower annual incomes.

Risk preferences: in Panel E of Table 6, estimates based on individuals' risk preferences are outlined. The total sample was divided into two sub-samples: one with lower risk preferences and another with higher risk preferences. The findings show that IMIS significantly raises the probability of return-to-hometown entrepreneurship by 0.61% (p < 0.01) for those with lower risk preferences. Conversely, the effect of IMIS on individuals with higher risk preferences, while positive, fails to attain statistical significance. This implies that IMIS exerts a more substantial incentive effect on migrant workers with lower risk preferences compared to those with higher risk preferences.

Social networks: the findings presented in Panel F of Table 6 demonstrate that the impact of IMIS on return-to-hometown entrepreneurship is contingent upon the magnitude of migrant

Age: according to the age classification criteria for middle-aged and older adults established by the World Health Organization (64–66), we divided the total sample into two sub-samples based on the median age of 45. As demonstrated in Panel B of Table 6, IMIS enhances the likelihood of migrant workers under 45 returning to their hometowns for entrepreneurial endeavors by 0.60% (p < 0.01). However, among

¹⁸ We also divide the total sample into two sub-samples based on the median age (40years) of our sample. The results reveal that the effects of IMIS are 0.0047 (p < 0.05) for individuals aged under 40 and 0.0043 (p < 0.1) for those aged over 40. Both of these effects are statistically significant. However, this does not affect our interpretation of the age heterogeneity results.

¹⁹ In China, compulsory education spans 9years, culminating with the completion of junior high school. This signifies that individuals, upon concluding compulsory education, have the option to voluntarily pursue further studies or enter the workforce. Therefore, our categorization of high and low education is based on the completion of compulsory education. Nonetheless, higher education also typically denotes the attainment of a bachelor's degree or higher, and thus, we conduct additional analysis using this criterion. The outcomes align closely with the results depicted in Table 5, panel c.

TABLE 6 Heterogeneity results.

Variables	(1)	(2)		
Panel A: Subgroups by gender				
	Female	Male		
IMIS	0.0030	0.0051**		
	(0.0025)	(0.0021)		
Observations	24,395	41,375		
Panel B: Subgroups by	age			
	≤45	>45		
	0.0060***	0.0025		
IMIS	(0.0020)	(0.0030)		
Observations	41,020	24,750		
Panel C: Subgroups by	educational level			
	Low	High		
IMIS	0.0050**	0.0026		
	(0.0020)	(0.0029)		
Observations	46,189	19,581		
Panel D: Subgroups by	annual income			
	Low	High		
DATE	-0.0015	0.0088***		
IMIS	(0.0027)	(0.0020)		
Observations	32,885	32,885		
Panel E: Subgroups by	risk preferences			
	Low	High		
IMIS	0.0061***	0.0005		
	(0.0018)	(0.0034)		
Observations	48,299	17,471		
Panel F: Subgroups by social networks				
	Small	Large		
IMIS	0.0033**	0.0057**		
	(0.0020)	(0.0026)		
Observations	32,885	32,885		
Control variables	Yes	Yes		

Control variables are the same as in Table 3. Standard errors are clustered at the individual level. ***, **, and * denote significance levels of 1, 5, and 10%, respectively.

workers' social networks, assessed through *renqing* expenditures.²⁰ Specifically, migrant workers with larger social networks are significantly more inclined to return to their hometowns for entrepreneurship following IMIS, with a significant increase of 0.57% (p < 0.05). On the other hand, the influence of IMIS on those with smaller social networks is less conspicuous, registering a change of

0.0033, which is not statistically significant—a circumstance potentially attributed to the smaller sample size. These results indicate that IMIS exerts a more potent motivating effect on migrant workers with expansive social networks compared to those with more limited networks.

5.5 Moderating effect of MEI

In 2015, "Mass Entrepreneurship and Innovation"(MEI) was incorporated into the Chinese government's work report. Subsequently, in 2016, China issued several policy documents aimed at promoting innovation and entrepreneurship among its residents. Therefore, the implementation of MEI occurred within the study period of this paper. A natural question arises: will the primary effect of IMIS in promoting return-to-hometown entrepreneurship among migrant workers be affected by MEI? To control for the potential contamination effects of MEI and examine how MEI might impact our results, we introduce the implementation year of MEI as a moderating variable and establish the following moderation effect model:

$$RE_{ipt} = \beta_0 + \beta_1 IMIS_{ipt} + \beta_2 MEI + \beta_3 IMIS_{ipt} \times MEI + \gamma X_{ipt} + \tau_p + \omega_t + \varepsilon_{ipt}$$
(4)

where the *MEI* is the moderator variable, a dummy variable which takes the value of 1 if the survey waves occurred in 2013 and 2015, and takes the value of 0 if the survey waves occurred in 2017 and 2019. β_1 represents the impact of IMIS on migrant workers' returnto-hometown entrepreneurship, controlling for the influence of MEI. β_2 signifies the influence of MEI on migrant workers' returnto-hometown entrepreneurship, controlling for the effect of IMIS. β_3 denotes the interactive effect between IMIS and MEI on migrant workers' return-to-hometown entrepreneurship. In other words, it elucidates how MEI moderates the impact of IMIS on migrant workers' return-to-hometown entrepreneurship. The settings for other variables are the same as those in Equation (1).

In Table 7, based on Equation (4), the moderating effect of MEI is analyzed. Column (3) indicates a positive but non-significant coefficient for IMIS (p > 0.1), suggesting that the impact of the IMIS on migrant workers' entrepreneurship becomes less significant when considering MEI. Conversely, the coefficient of MEI is significant (p < 0.05), underscoring a stronger influence of MEI on entrepreneurship under the control of IMIS. Notably, the interaction between IMIS and MEI shows an increased coefficient of 0.0069 (p < 0.01), compared to IMIS's coefficient of 0.0044 (p < 0.01) in Column (4) of Table 3. This implies that MEI amplifies the effect of IMIS in encouraging migrant workers to start businesses in their hometowns. The result is unsurprising, as IMIS meets the basic healthcare needs of migrant workers, providing a more stable medical safeguard, thereby reducing entrepreneurial risks. Concurrently, the MEI policy offers substantial entrepreneurial opportunities and support. These two policies likely create synergistic and complementary effects in different aspects, fostering a more conducive environment for entrepreneurship. This invigorates migrant workers' enthusiasm for returning to their hometowns to start businesses.

²⁰ According to the contents of the questionnaire, expenditures related to social interactions include discretionary spending during holidays (e.g., the Spring Festival and Mid-Autumn Festival), due to celebratory events (e.g., weddings and funerals), for education, for healthcare, for living, and for other purposes.

TABLE 7 Moderating effect of MEI.

Variables	(1)	(2)	(3)
IMIS	-0.0016 (0.0015)	-0.0010 (0.0015)	0.0017 (0.0022)
MEI	0.0014 (0.0014)	0.0043*** (0.0015)	0.0035** (0.0018)
IMIS×MEI	0.0158*** (0.0005)	0.0182*** (0.0012)	0.0069*** (0.0013)
Control variables	No	Yes	Yes
Province fixed effects	No	No	Yes
Observations	65,770	65,770	65,770
R ²	0.0040	0.0086	0.0122

Control variables are included at individual and household levels as shown in Table 3. Standard errors are clustered at the individual level. ***, **, and * denote significance levels of 1, 5, and 10%, respectively.

6 Discussion

The primary result reveals that IMIS significantly increases the probability of migrant workers engaging in return-to-hometown entrepreneurship by 0.44%, which still holds after a series of robustness tests. According to the "Statistical Communique of the People's Republic of China on National Economic and Social Development for the Year 2022,"21 in 2022 there were approximately 171.9 million migrant workers in China. According to our calculations, IMIS has the potential to facilitate the return of 756,000 individuals to rural areas for entrepreneurial pursuits. It is essential to note that this is a relatively conservative estimate, as it may be affected by bias due to the presence of negative weights in the DID estimation. Upon addressing this issue, the magnitude of the effect could potentially escalate to around 1%. This increase can be attributed to the improvements in welfare benefits compared to the previous NCRMS, particularly the expansion of medical insurance generosity and higher reimbursement rates. With the reduction in disparities between urban and rural medical coverage, some migrant workers may be more inclined to return to rural areas due to lower living costs, both for residence and entrepreneurship, thereby contributing to the return of the original rural workforce (68). While prior research has not specifically assessed the effect of IMIS on the return-to-hometown entrepreneurship of migrant workers, certain studies have explored the influence of health insurance on the employment, entrepreneurship or pullback effect of migrant workers (3, 68, 69), and their findings are consistent with this paper.

The results of the heterogeneity show that IMIS has a higher intensity of impact on male migrant workers returning to their hometowns for entrepreneurial endeavors. Possible explanations for this divergence include, firstly, within traditional Chinese cultural norms, males typically shoulder the role of primary breadwinners, whereas women are predominantly tasked with family care responsibilities. This dynamic may pose additional challenges for women contemplating a return to entrepreneurship in their hometowns (3). Secondly, gender inequalities in the labor market, which may result in greater challenges for females in accessing necessary resources for entrepreneurship, including funding, technical support, and networks (70). Thirdly, gender differences in industry or professional preferences in the context of hometown entrepreneurial ventures may also contribute to their varied responses to IMIS (71). For instance, females may lean toward social enterprises or community-oriented entrepreneurship, whereas males may focus more on commercial gains and expansion.

In terms of age, there is an increased probability of younger individuals engaging in hometown entrepreneurship after the implementation of IMIS. In the context of China's rural labor shortage, particularly among the youth, this trend undoubtedly contributing to alleviating the issue of rural hollowing. The reasons for this phenomenon can be reasonably explained by the Life Cycle Theory and Human Capital Theory (72, 73). The former suggests that younger individuals possess greater flexibility in career choices and geographic mobility, while the latter emphasizes the investment of young people in education and skills, which may predispose them to capitalize on opportunities for returning to their hometowns to start businesses.

Concerning educational level, migrant workers with lower education levels typically have fewer savings and limited social security (74). These circumstances can make them more apprehensive about potential medical expenses, which in turn could pose a significant obstacle to their entrepreneurial aspirations when returning home. The implementation of IMIS has expanded healthcare coverage, providing them with the confidence required for participation in rural entrepreneurial activities, thus they may be more inclined to return to rural areas to seek or create job opportunities (44). Furthermore, migrant workers with higher incomes typically possess increased savings and financial resources, providing them with a financial buffer that enables the initiation and sustenance of entrepreneurial endeavors (75). The implementation of IMIS unlocks precautionary savings that can be redirected toward investments in entrepreneurship, significantly increasing their likelihood of returning home for business ventures.

We also creatively analyze heterogeneity in risk preferences and social networks. Due to the different risk preferences of individual migrant workers, the impact of IMIS on their return to hometowns to become an entrepreneur is not the same. One plausible explanation is that IMIS, by alleviating concerns regarding health risks and medical expenses (76), provides a degree of economic security (77). This observation is consistent with Risk Aversion Theory, positing that risk-averse individuals are inclined to favor options that minimize uncertainty and potential losses. For individuals with lower risk preferences, they may be more inclined to prioritize their health, economic security, and family well-being (22). As a result, they show more caution before making decisions (78), which lead to their hesitation to embrace the risks of entrepreneurship prior to IMIS, and in turn, they chose to return to their hometowns for entrepreneurship.

We have found that social networks are instrumental in entrepreneurship (79). This observation can be elucidated through the framework of social capital theory, emphasizing the crucial role of social networks in furnishing resources, information, and social support (80). Additionally, the size of one's social network may potentially correlate with entrepreneurial motivation. For example, a larger network implies greater social support (capital, land, equipment,

²¹ https://www.stats.gov.cn/xxgk/sjfb/zxfb2020/202302/

t20230228_1919001.html?egid=fa9f347f0008349f00000002645671cd

or other forms of support) and encouragement (relatives, friends, fellow townspeople, etc.), possibly leading to more opportunities and ideas for entrepreneurship. This, in turn, is vital for stimulating the entrepreneurial interest and enthusiasm of migrant workers, as well as for reducing the perception of entrepreneurial risks and enhancing confidence in their entrepreneurial pursuits (81). However, it is important to note that our proxy variable for social networks, renging expenditures, may not fully encapsulate the complexities of these relationships. The formation and sustenance of social networks are multifaceted, involving elements like family ties, friendships, professional connections, and participation in social activities (82). Renging expenditures represent just one dimension of these interactions, mainly indicative of the level of friendly interactions, cooperation, and mutual assistance. Consequently, while these results are enlightening, they ought to be interpreted cautiously due to the inherent limitations of using renging expenditures as a surrogate for the breadth of social networks (83).

The results pertaining to the moderating effect reveal a crucial fact: while the IMIS exerts a positive impact on the entrepreneurial decisions of migrant workers, this influence is significantly amplified when combined with MEI. In other words, there exists a distinct synergistic effect between medical security and entrepreneurship support policies. This indicates that to effectively promote entrepreneurship among returning migrant workers, a solitary policy approach may be insufficient in yielding substantial outcomes, and the implementation of comprehensive policy measures is crucial. This conclusion aligns with established theoretical frameworks, suggesting that an integrated policy approach can more effectively stimulate and support the actions of the target group.

We acknowledge potential limitations in our study: Firstly, due to data limitations, we could only analyze information from four survey waves, limiting our ability to capture comprehensive pre- and posttrends. Nevertheless, we believe that our results still offer valuable insights and implications. Secondly, while we have presented empirical evidence of IMIS's role in promoting the return of rural labor for entrepreneurship, the ultimate outcomes of migrant workers' entrepreneurship remain uncertain, necessitating further exploration in future studies. Thirdly, due to data limitations, we are unable to explore the potential mechanisms behind the results of this study. However, we believe that a reduction in healthcare burdens and an increase in risk-taking capacity could be potential mechanisms. On the one hand, after the implementation of IMIS, migrant workers can receive higher reimbursements when they fall ill due to the improved reimbursement rates. On the other hand, the reduced economic risks associated with illness may make migrant workers more willing to take entrepreneurial risks. Nevertheless, these mechanisms would require further investigation in future study.

7 Conclusion and policy implications

In this study, we use panel data from CHFS from 2013 to 2019 to empirically test the effect of the IMIS on migrant workers' return-tohometown entrepreneurship by employing a time-varying DID method. The main conclusions are as follows: Firstly, we find that the IMIS significantly increases the likelihood of migrant workers returning to their hometowns for entrepreneurship by 0.44%, which remains valid after a series of robustness checks, including DIDM, endogeneity, alternative models, and placebo tests. Secondly, this effect is more pronounced for those who are male and with lower educational levels, higher income, larger social networks, and lower risk preferences. Thirdly, the policy synergy of MEI and IMIS can yield greater effects on the return of migrant workers to their hometowns for entrepreneurship.

This paper provides the following policy implications: Firstly, policymakers should concentrate on further enhancing the healthcare support system in rural areas. This involves a continuous extension of medical insurance coverage and a judicious augmentation of welfare benefits for migrant workers, intending to alleviate the financial strain associated with potential medical expenditures during the entrepreneurial journey and foster heightened entrepreneurial motivation. Secondly, the government and society should pay attention to the medical protection of the key groups of rural migrant workers, and formulate different details of the medical insurance system for different situations, so that the entrepreneurial environment of the rural migrant workers can be more effectively protected. In addition, it is necessary to guide and organize migrant workers to actively participate in urban and rural residents' basic medical insurance, so as to make up for the short board of medical insurance for migrant workers and effectively improve their medical insurance treatment. Thirdly, it is recommended that relevant authorities bolster risk education initiatives for rural migrant workers, facilitating their rational understanding of risks and augmenting their capacity to manage and mitigate them. Furthermore, it is crucial to cultivate collaboration between governments and social organizations at various levels, aiming to construct social network "bridges" for rural migrant workers aspiring to return to their hometowns and embark on entrepreneurial ventures. Additionally, it is essential to encourage migrant workers to actively participate in various social activities, which will help to broaden their social network resources and ultimately promote their entrepreneurship in their respective hometowns. Lastly, the optimization of a system should take into account the systemic and holistic nature (84). Policymakers, in their drive to promote rural revitalization, should consider medical security and entrepreneurship support as an integrated whole to achieve the maximum policy effectiveness.

Our research findings also carry implications for other developing countries with similar welfare program improvements and efforts aimed at rebalancing urban and rural population migration structures. Over the past decade, China has continuously enhanced the welfare provisions within its rural labor market plan. Consequently, we anticipate that the impact of this program on labor supply has become increasingly prominent in recent years. However, it is essential to note that the expected effects of certain welfare programs have yet to be rigorously analyzed. As such, it is crucial that future research seeks to validate the individual and combined effects of these new policies using updated and more robust datasets. This will enable a more comprehensive understanding of the influence of welfare improvements on labor dynamics and the broader socio-economic landscape in developing countries.

Data availability statement

Publicly available datasets were analyzed in this study. This data can be found at: China Household Finance Survey (CHFS), https:// chfs.swufe.edu.cn/index.htm.

Author contributions

PH: Conceptualization, Funding acquisition, Resources, Writing – original draft. ZS: Data curation, Formal analysis, Methodology, Software, Supervision, Validation, Visualization, Writing – original draft, Writing – review & editing. LL: Writing – original draft. JL: Supervision, Validation, Visualization, Writing – review & editing.

Funding

The author(s) declare financial support was received for the research, authorship, and/or publication of this article. This work was supported by Shandong Province Higher Education Institutions Youth Innovation and Technology Support Program (No. 2022RW044).

References

1. Zhang R, Jiang G, Zhang Q. Does urbanization always Lead to rural hollowing? Assessing the Spatio-temporal variations in this relationship at the county level in China 2000–2015. *J Clean Prod.* (2019) 220:9–22. doi: 10.1016/j.jclepro.2019.02.148

2. Wang Y, Wang S. Issues on aging of population and poverty of aged people in rural China--relating research on migration influences. *J China Agricultural University (Social Sciences)*. (2014) 31:108–20. doi: 10.13240/j.cnki.caujsse.2014.01.009

3. Wang Y, Shi J, Yao Y, Sun W. The impact of health insurance on job location choice: evidence from rural China. *J Comp Econ.* (2022) 50:569–83. doi: 10.1016/j. jce.2022.01.001

4. Enchautegui ME. Welfare payments and other economic determinants of female migration. J Labor Econ. (1997) 15:529–54. doi: 10.1086/209871

5. McKinnish T. Welfare-induced migration at state Borders: new evidence from Micro-data. J Public Econ. (2007) 91:437–50. doi: 10.1016/j.jpubeco.2006.09.002

 Giulietti C, Wahba J. Welfare migration In: KF Zimmermann and AF Constant, editors. *International handbook on the economics of migration*. UK: Edward Elgar Publishing (2013).

7. de Giorgi G, Pellizzari M. Welfare migration in Europe. Labour Econ. (2009) 16:353–63. doi: 10.1016/j.labeco.2009.01.005

8. Buchmueller TC, Valletta RG. The effects of employer-provided health insurance on worker mobility. *ILR Rev.* (1996) 49:439–55. doi: 10.1177/001979399604900304

9. Cooper PF, Monheit AC. Does employment-related health insurance inhibit job mobility? *Inquiry*. (1993) 30:400–16.

10. Gruber J, Madrian BC. Health insurance, labor supply, and job mobility: A critical review of the literature. Health policy and the uninsured. Washington D.C.: Urban Institute Press (2004).

11. Madrian BC. Employment-based health insurance and job mobility: is there evidence of job-lock? *Q J Econ*. (1994) 109:27–54. doi: 10.2307/2118427

12. Stroupe KT, Kinney ED, Kniesner TJ. Chronic illness and health insurance-related job lock. *J Policy Analysis and Manag: The J Assoc Public Policy Analysis and Manag.* (2001) 20:525–44. doi: 10.1002/pam.1006

13. Shi X. Locked out? China's health insurance scheme and internal migration. Labour Econ. (2020) 67:101931. doi: 10.1016/j.labeco.2020.101931

14. Feng J, Chen X, Hu B. The medical insurance based on efficiency and equity: evidence from the integration of the National Basic Medical Insurance Program for urban and rural residents. *Econ Res J.* (2022) 57:154–72.

15. Li Y, Liu N, Chen H, Shen B. Has the integrated medical insurance relieved the overwork of migrant workers? *Chinese Rural Econ.* (2022) 7:124–44.

16. Liu Q, He S, Tian P, Qin S. How does the urban-rural integrated medical insurance affect the well-being of rural elderly? -- evidence from CHARLS data. *Theory and Practice of Finance and Econ.* (2022) 4:43–50. doi: 10.16339/j.cnki.hdxbcjb.2022.04.006

17. Liu P, Guo W, Liu H, Hua W, Xiong L. The integration of urban and rural medical insurance to reduce the rural medical burden in China: a case study of a county in Baoji City. *BMC Health Serv Res.* (2018) 18:1–9. doi: 10.1186/s12913-018-3611-y

18. Huo J, Hu M, Li S. The impact of urban-rural medical insurance integration on medical impoverishment: evidence from China. *Int J Equity Health.* (2023) 22:245. doi: 10.1186/s12939-023-02063-6

19. Ma C, Zhao GC, Gu H. The effect of urban-rural integrated medical insurance system on rural residents' health care behavior. *Statistical Res.* (2016) 33:78–85. doi: 10.19343/j.cnki.11-1302/c.2016.04.011

Conflict of interest

The authors declare that the research was conducted in the absence of any commercial or financial relationships that could be construed as a potential conflict of interest.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

20. Zhou Q, He Q, Eggleston , Liu GG. Urban-rural health insurance integration in China: impact on health care utilization, financial risk protection, and health status. *Appl Econ.* (2022) 54:2491–509. doi: 10.1080/00036846.2021.1998323

21. Li C, Tang C, Wang H. Effects of health insurance integration on health care utilization and its equity among the mid-aged and elderly: evidence from China. *Int J Equity Health.* (2019) 18:1–12. doi: 10.1186/s12939-019-1068-1

22. Ma W, Li H. From new rural cooperative medical insurance to basic medical Insurance for Urban and Rural Eesidents: an analysis of policy effects of the urban-rural medical insurance coordination system. *China Econ Stud.* (2021) 4:146–57. doi: 10.19365/j.issn10004181.2021.04.11

23. Holzmann R, Koettl J. Portability of pension, health, and other social benefits: facts, concepts, and issues. *CESifo Econ Stud.* (2015) 61:377–415. doi: 10.1093/cesifo/ift017

24. Ma C, Gu H, Sun X. Does the pooling of urban and rural medical insurance contribute to the psychological integration of rural floating population? *China Rural Observa.* (2017) 2:41–53.

25. Ma C, Gu H, Sun X. The effect of medical insurance pooling mode on the utilization of medical services and the substantial equity of health in urban and rural residents. *J Pub Administr.* (2017) 14:97–109.

26. Chang X, Su Q, Zhou C. The effects of urban-rural integrated medical insurance system on health. *Chin J Pop Sci.* (2018) 6:103–14+128.

27. Ren Y, Zhou Z, Cao D, Ma BH, Shen C, Lai S, et al. Did the integrated urban and rural resident basic medical insurance improve benefit equity in China? *Value Health.* (2022) 25:1548–58. doi: 10.1016/j.jval.2022.03.007

28. Xu X. Urban-rural integrated medical insurance system and rural migrants' healthcare service utilization: analysis based on the 2017 CMDS data. *Chinese J Health Policy*. (2022) 3:53–60. doi: 10.3969/j.issn.1674-2982.2022.03.008

29. Lewis A. Economic development with unlimited supplies of labor. *Manch Sch.* (1954) 22:139–91. doi: 10.1111/j.1467-9957.1954.tb00021.x

30. Todaro MP. A model of labor migration and urban unemployment in less developed countries. *Am Econ Rev.* (1969) 59:138–48. doi: 10.2307/1811100

31. Kong X, Chen X. How resource endowment differences influence entrepreneurship of migrant workers: evidence from CHIP2013 data. *Industrial Econ Rev.* (2018) 9:112–21. doi: 10.14007/j.cnki.cjpl.2018.05.009

32. Wang Y, Song S, Peng Y, Zhou A, Yang Y. Research on the influencing factors of rural migrant workers' entrepreneurship willingness based on interactive determinism. *Journal of Minzu University of China (Philosophy and Social Sciences Edition)*. (2020) 47:120–9. doi: 10.15970/j.cnki.1005-8575.2020.03.013

33. Li C, Liu X, Wang S. Credit constraints, social networks and entrepreneurial implementation of new-generation migrant farmers. *J Agro-Forestry Econ Manag.* (2019) 18:607–17. doi: 10.16195/j.cnki.cn36-1328/f.2019.05.65

34. Luo M. Individual characteristics, resource acquisition and Farmers' entrepreneurship: an empirical analysis based on questionnaire survey data in selected regions of Guangdong Province. *China Rural Survey*. (2012) 2:11–9.

35. Zhu H, Kang L. The financial Environent, policy support and Peasants' entrepreneurial willingness. *China Rural Survey*. (2013) 5:24–33+95–96.

36. Zhu MF. Analysis of factors influencing Farmers' entrepreneurial behavior--taking Hangzhou, Zhejiang as an example. *Chinese Rural Economy*. (2010) 3:25–34.

37. Jiang J, Qian W, Guo H. Social networks, prior experience and Farmers' entrepreneurial decisions. *J Agrotechnical Econ.* (2014) 2:17–25. doi: 10.13246/j.cnki. jae.2014.02.002

38. Wang L., Wang A., FitzGerald G., Si L., Jiang Q., Ye D. Who benefited from the new rural cooperative medical system in China? A case study on Anhui Province. *BMC Health Serv Res.* (2016) 16:1–8. doi: 10.1186/s12913-016-1441-3

39. Zhang Z, Chen H, Li Z. Does health insurance improve health—a health effect evaluation of the new cooperative medical scheme in rural China. *Social Security Stud.* (2015) 4:28–35.

40. Chen J, Yu H, Dong H. Effect of the new rural cooperative medical system on farmers' medical service needs and utilization in Ningbo, China. *BMC Health Serv Res.* (2016) 16:593. doi: 10.1186/s12913-016-1842-3

41. Chu L. Improvement of Farmers' health Insurance Services in the Context of urban and rural Residents' health insurance integration and its optimization strategy research. *Theory J.* (2022) 3:141–9. doi: 10.14110/j.cnki.cn-37-1059/d.2022.03.008

42. Liu J. How to realize universal health Insurance in China? --a study on the influencing factors of the development of social health insurance system. *Comp Econ Soc Syst.* (2010) 2:15–22.

43. Zhong H, Wang Z. "Single system, single-standard" vs. "single system, multistandard"—the impact of the financing method of Chinese urban-rural integrated medical insurance system on the rate of joining Insurance for Middle-Aged and Elderly People. *Sustainability*. (2021) 14:274. doi: 10.3390/su14010274

44. Chang X, Su Q, Xie H. The influence of urban and rural medical insurance system on the medical burden of residents-based on the difference in urban and rural areas. *J Guizhou University of Finance and Econ.* (2021) 2:80–9.

45. Fan XY, Liu KJ, Lou YT, Gao LQ, Dong HJ. Analysis on current status and differences of medical insurance policy for chronic and special diseases in medical insurance outpatients for urban and rural residents in Zhejiang Province. *China Health Insurance*. (2023) 7:85–90. doi: 10.19546/j.issn.1674-3830.2023.7.011

46. Qiu YL, Zhang PF. From "National Medical Insurance" to "fair medical insurance": current situation assessment and path analysis of the integration of medical insurance system for urban and rural residents in China. J Hebei University (Philosophy and Social Sci). (2019) 44:128–38.

47. Qi J, Zheng X, Cao P, Zhu L. The effect of e-commerce agribusiness clusters on farmers' migration decisions in China. *Agri.* (2019) 35:20–35. doi: 10.1002/agr.21586

48. Xie L. Employment of rural labor force in the 14th five-year plan period: situation outlook. Structure forecast and thought countermeasure. *Issues Agric Econ.* (2021) 3:28–39. doi: 10.13246/j.cnki.iae.2021.03.003

49. Xu Y, Xu H. "Robotization": technological upgrading and its impact. *Beijing Cult Rev.* (2019) 3:88–96.

50. Wang AL, Zheng HG, Zhang JJ, Shen LM. Comparison and enlightenment of policies and measures of returning to rural areas in South Korea and China. *World Agriculture*. (2021) 2:74–132. doi: 10.13856/j.cn11-1097/s.2021.02.009

51. Bao A, Pang G, Zeng G. Entrepreneurial effect of rural return migrants: evidence from China. *Front Psychol.* (2022) 13:1078199. doi: 10.3389/fpsyg.2022.1078199

52. Cao Z. The multiple causes and exceptional functions of migrant Workers' returning home to run business under new Normal economy. *Social Sci Guangdong*. (2019) 3:23–30+254.

53. Naminse EY, Zhuang J, Zhu F. The relation between entrepreneurship and rural poverty alleviation in China. *Manag Decis.* (2018) 57:2593–611. doi: 10.1108/MD-11-2017-1153

54. Wu X, Chen L, Ma L, Cai L, Li X. Return migration, rural household investment decision, and poverty alleviation: evidence from rural Guangdong, China. *Growth Chang*, (2022) 54:304–25. doi: 10.1111/GROW.12656

55. Beck T, Levine R, Levkov A. Big bad banks? The winners and Losers from Bank deregulation in the United States. *J Financ.* (2010) 65:1637–67. doi: 10.1111/j.1540-6261.2010.01589.x

56. Shi L, Tan Y, Wu H. Analysis of entrepreneurial behavior and entrepreneurial willingness of returning rural migrant workers. *China Rural Survey*. (2010) 5:25–37.

57. Guo Y, Wang C. The new cooperative medical system and entrepreneurship. *China Econ Q.* (2016) 15:1463–82. doi: 10.13821/j.cnki.ceq.2016.03.08

58. Chen Y, Zhang B. The influence of social security on the decision of migrant workers' mobility: an empirical study based on the theory of push-pull. *Issues in Agricul Economy*. (2018) 10:132–40. doi: 10.13246/j.cnki.iae.20181013.002

59. Lu SF, Hong JT. Rural E-commerce construction and regional coordinated development: evidence from China's E-commerce pilots rural areas. *Econ Rev.* (2023) 5:71–88. doi: 10.19361/j.er.2023.05.05

60. Zhen H, Jia N. Does the use of robots exacerbate the return of rural migrate labor force? *Contemporary Finance & Economics.* (2022) 12:3–15. doi: 10.13676/j.cnki. cn36-1030/f.2022.12.002

61. de Chaisemartin C, D'Haultfœuille X. Two-way fixed effects estimators with heterogeneous treatment effects. *Am Econ Rev.* (2020) 110:2964–96. doi: 10.1257/aer.20181169

62. de Chaisemartin C, D'Haultfœuille X. Difference-in-differences estimators of intertemporal treatment effects. *National Bureau of economic Res.* (2022):39.

StataCorp. stata statistical software: Release17. StataCorp LLC. (2021).

64. Fan HL, Yang JL, Zhang XH. The impact of social cultural organizations on rural residents' happiness:an analysis based on data analysis from CLDS survey. *China Rural Survey*. (2022) 4:170–84.

65. Lyu D, Bai F, Zhang JT. Agricultural land property rights and interests and farmers' consumption: an empirical analysis on the consumption of urban-rural integration development households. *China Soft Sci.* (2023) 6:167–79.

66. Wang HL, Zhu XS. The influencing factors of health inequality on the middle-aged and elderly. Chinese journal of population. *Science*. (2018) 3:109–28.

67. Tamvada J, Shrivastava M, Mishra T. Education, social identity and selfemployment over time: evidence from a developing country. *Small Bus Econ.* (2022) 59:1449–68. doi: 10.1007/s11187-021-00583-5

68. Shi X, Wang C, Zhong T. Can public medical insurance promote rural entrepreneurship? *Evidence From China Applied Economics*. (2021) 37:4292–309. doi: 10.1080/00036846.2021.1899118

69. Qin L, Chen CP, Wang W, Chen H. How migrants get integrated in urban China - the impact of health insurance. *Soc Sci Med.* (2021) 272:113700. doi: 10.1016/j. socscimed.2021.113700

70. Khalid S, Dixon S, Vijayasingham L. The gender responsiveness of social entrepreneurship in health–a review of initiatives by Ashoka fellows. *Soc Sci Med.* (2022) 293:114665. doi: 10.1016/j.socscimed.2021.114665

71. Li J, Zhang X, Xie J, Yang J. Review of the impact of gender difference on entrepreneurship. *Sci Technol Progress and Policy*. (2009) 26:190–4.

72. Gielnik MM, Zacher H, Wang M. Age in the entrepreneurial process: the role of future time perspective and prior entrepreneurial experience. *J Appl Psychol.* (2018) 103:1067. doi: 10.1037/apl0000322

73. Thorgren S, Sirén C, Nordström C, Wincent J. Hybrid entrepreneurs' second-step choice: the nonlinear relationship between age and intention to enter full-time entrepreneurship. J Bus Ventur Insights. (2016) 5:14–8. doi: 10.1016/j.jbvi.2015.12.001

74. Revoredo C, Morisset J. Savings and education: A life-cycle model applied to a panel of 74 countries. (1999) doi: 10.1596/1813-9450-1504

75. Pisure BL, Thombre BM, Sidam VN. Relationship between personal, socioeconomic and psychological characteristics of dairy farmers and their entrepreneurial behavior. *Int J Farm Sci.* (2014) 4:264–71.

76. Wang Y, Li Y, Zhao R, Yao R, Li W, Wu R. The mechanism of the influence of returning migrant Workers' rural tourism entrepreneurial motivation on entrepreneurial well-being: an empirical study based on chain mediation effect. *J Central China Normal University (Natural Sci)*. (2023) 9:1–25.

77. Wang P. How to cultivate IMIS farmers. People's Tribune. (2019) 4:78-9.

78. Zhou G, Tan H, Li L. Dose migration experience promote entrepreneurship in rural China. *China Econ Q*. (2017) 16:793–814. doi: 10.13821/j.cnki.ceq.2017.01.15

79. Witt P. Entrepreneurs' networks and the success of start-ups. *Entrep Reg Dev.* (2004) 5:391–412. doi: 10.1080/0898562042000188423

80. Hu W, Xu Y, Zhao F, Chen Y. Entrepreneurial passion and entrepreneurial success—the role of psychological capital and entrepreneurial policy support. *Front Psychol.* (2022) 13:792066. doi: 10.3389/fpsyg.2022.792066

81. Sheng Y, Shang J, Zheng H. The influence of clan culture on urban entrepreneurship of migrants. *Population Res.* (2023) 4:114–28.

82. Hassan HM, Igel B, Shamsuddoha M. Entrepreneurship education and social entrepreneurial intentions: the mediating effects of entrepreneurial social network. *Front Psychol.* (2022) 13:860273. doi: 10.3389/fpsyg.2022.860273

83. Wen G, He HM, Qin Y. The impact of psychological capital and social capital on the entrepreneurial performance of returning farmers—based on the sample data of Sichuan Province. *J Southwest University (Natural Sci Edition)*. (2023) 45:45–56. doi: 10.13718/j.cnki.xdzk.2023.07.004

84. Yu X, Hu Q, Shen Y. The mechanism and solution of the puzzle of "retirement-medical service fluctuation": theoretical and empirical analysis from systematic connection perspective. *J Manag World*. (2023) 39:94–115. doi: 10.19744/j.cnki.11-1235/f.2023.0121